Author's response to reviews

Title: Lemierres syndrome and genetic polymorphisms : Case Report

Authors:

Jean-michel Constantin (jmconstantin@chu-clermontferrand.fr)
Jean-paul Mira (jean-paul.mira@cch.ap-hop-paris.fr)
Renaud Guerin (rguerin@chu-clermontferrand.fr)
Sophie Cayot-constantin (scayot@chu-clermontferrand.fr)
Olivier Lesens (olesens@chu-clermontferrand.fr)
Florence Gourdon (fgourdon@chu-clermontferrand.fr)
Jean-pierre Romaszko (jpromaszko@chu-clermontferrand.fr)
Philippe Linval (philippe.linval@wanadoo.fr)
Henri Laurichesse (hlaurichesse@chu-clermontferrand.fr)
Jean-etienne Bazin (jebazin@chu-clermontferrand.fr)

Version: 5 Date: 14 June 2006

Author's response to reviews:

As requested by the reviewer Dr Gordon, we added in the Table the rs number, of the SNP, which defined precisely the studied polymorphism and act as an international ID for each of them. For the variant localized in the promotor par of the tissu factor gene, we were unable to find the rs number but we gave one of the publication related to this specific polymorphism.
As requested by the reviewer Dr Opal, all the typos have been corrected.